Author's response to reviews

Title: The effects of EPA and DHA enriched fish oil on nutritional and immunological markers of treatment naïve breast cancer patients: A randomized double-blind controlled trial

Authors:

Elemárcia Paixão (e.paixao@yahoo.com.br)
Ana Carolina Oliveira (anynha.carol@yahoo.com.br)
Nathalia Pizato (pizatonat@unb.br)
Maria Imaculada Muniz-Junqueira (mimjunqueir@unb.br)
Kelly Grace Magalhães (kellymagalhaes@unb.br)
Eduardo Nakano (nakano@unb.br)
Marina Ito (marina@unb.br)

Version: 2 Date: 29 Sep 2017

Author’s response to reviews:

Reviewer #2:

The authors have addressed most of the concerns raised previously however in the interest of clarity and transparency in reporting outcomes I do think that the lipid and glucose results should be included in text regardless of the lack of significance so the authors are asked to add these details.

We added a new table where we present the biochemical parameters evaluated. Please, see page 11, line 282 and table 4.